3-month versus 6-month adjuvant chemotherapy for patients with high-risk stage II and III colorectal cancer: 3-year follow-up of the SCOT non-inferiority RCT

Timothy Iveson, 1* Kathleen A Boyd, 2 Rachel S Kerr,³ Jose Robles-Zurita,² Mark P Saunders,⁴ Andrew H Briggs,² Jim Cassidy,⁵ Niels Henrik Hollander, ⁶ Josep Tabernero, ⁷ Andrew Haydon,⁸ Bengt Glimelius,⁹ Andrea Harkin,⁵ Karen Allan,⁵ John McQueen,⁵ Sarah Pearson, 10 Ashita Waterston, 11 Louise Medley, 12 Charles Wilson, 13 Richard Ellis, 14 Sharadah Essapen, 15 Amandeep S Dhadda, 16 Mark Harrison, 17 Stephen Falk, 18 Sherif Raouf, 19 Charlotte Rees, ¹ Rene K Olesen, ²⁰ David Propper, ²¹ John Bridgewater,²² Ashraf Azzabi,²³ David Farrugia,²⁴ Andrew Webb,²⁵ David Cunningham,²⁶ Tamas Hickish,²⁷ Andrew Weaver, 28 Simon Gollins, 29 Harpreet Wasan³⁰ and James Paul⁴

¹Southampton University Hospital NHS Foundation Trust, Southampton, UK

²Institute of Health and Wellbeing, University of Glasgow, Glasgow, UK

³Department of Oncology, University of Oxford, Oxford, UK

⁴The Christie Hospital NHS Foundation Trust, Manchester, UK

⁵Cancer Research UK Clinical Trials Unit, Institute of Cancer Sciences, University of Glasgow, Glasgow, UK

⁶Department of Oncology and Palliative Care, Zealand University Hospital, Naestved, Denmark

⁷Vall d'Hebron University Hospital and Institute of Oncology, Universitat Autònoma de Barcelona, Barcelona, Spain

⁸Australasian Gastro-Intestinal Trials Group, Camperdown, NSW, Australia ⁹University of Uppsala, Uppsala, Sweden

¹⁰Oncology Clinical Trials Office, Department of Oncology, University of Oxford, Oxford, UK

¹¹Beatson West of Scotland Cancer Centre, Glasgow, UK

- ¹²Royal United Hospital, Bath, UK
- ¹³Addenbrooke's Hospital, Cambridge, UK
- ¹⁴Royal Cornwall Hospitals NHS Trust, Cornwall, UK
- ¹⁵St Luke's Cancer Centre, Royal Surrey County Hospital NHS Foundation Trust, Guildford, UK
- ¹⁶Castle Hill Hospital, Hull, UK
- ¹⁷Mount Vernon Cancer Centre, Northwood, UK
- ¹⁸Bristol Cancer Institute, Bristol, UK
- ¹⁹Barking Havering and Redbridge University Hospital NHS Trust, Barking, UK
- ²⁰Department of Oncology, Aarhus University Hospital, Aarhus, Denmark
- ²¹Barts Cancer Institute, Queen Mary University of London, London, UK
- ²²Department of Oncology, University College London, London, UK
- ²³Newcastle upon Tyne Hospitals NHS Foundation Trust, Newcastle upon Tyne, UK
- ²⁴Gloucestershire Oncology Centre, Cheltenham General Hospital, UK
- ²⁵Brighton and Sussex University Hospital Trust, Brighton, UK
- ²⁶Royal Marsden NHS Foundation Trust, London, UK
- ²⁷Poole Hospital NHS Foundation Trust, Poole, UK
- ²⁸Department of Oncology, Oxford University Hospitals Foundation Trust, Oxford, UK
- ²⁹North Wales Cancer Treatment Centre, Rhyl, UK
- ³⁰Hammersmith Hospital, Imperial College Healthcare NHS Trust, London, UK

Declared competing interests of authors: Timothy Iveson reports honoraria from Amgen Inc. (Thousand Oaks, CA, USA), Bayer AG (Leverkusen, Germany), Bristol-Myers Squibb (New York, NY, USA), Celgene Corporation (Summit, NJ, USA), Pierre-Fabre (Paris, France), Roche (Roche Holding AG, Basel, Switzerland) and Servier (Laboratories Servier, Suresnes, France). Kathleen A Boyd reports grants from the Medical Research Council during the conduct of the study. Mark P Saunders reports personal fees from Servier, Amgen, Merck (Merck and Co., Kenilworth, New Jersey, USA), Eisai (Eisai Co., Ltd., Tokyo, Japan) and Roche outside the submitted work. Jim Cassidy reports grants from the Medical Research Council during the conduct of the study and is currently an employee of Celgene Corporation. Josep Tabernero reports personal fees from Array Biopharma (Boulder, CO, USA), AstraZeneca (Cambridge, UK), Bayer AG (Leverkusen, Germany), BeiGene (Beijing, China), Boehringer Ingelheim (Ingelheim am Rhein, Germany), Chugai (Chugai Pharmaceutical Co., Tokyo, Japan), Genentech, Inc. (South San Francisco, CA, USA), Genmab A/S (Copenhagen, Denmark), Halozyme (Halozyme Therapeutics, San Diego, CA, USA), Imugene Limited (Sydney, NSW, Australia), Inflection Biosciences Limited (Blackrock, Dublin), Ipsen (Paris, France), Kura Oncology (San Diego, CA, USA), Eli Lilly and Company (Indianapolis, IN, USA), Merck, Menarini (The Menarini Group, Florence, Italy), Merck Serono (Rockland, MA, USA), Merrimack Pharmaceuticals (MA, USA), Merus (Utrecht, the Netherlands), Molecular Partners (Molecular Partners AG, Zurich, Switzerland), Novartis (Novartis International AG, Basel, Switzerland), Peptomyc, Pfizer Inc. (New York, NY, USA), Pharmacyclics (Pharmacyclics LLC, Sunnyvale, CA, USA), ProteoDesign SL (Barcelona, Spain), Rafael Pharmaceuticals (Stony Brook, NY, USA), F. Hoffmann-La Roche Ltd, Sanofi (Sanofi S. A., Paris, France), Seattle Genetics (Bothwell, WA, USA), Servier, Symphogen (Symphogen A/S, Ballerup, Denmark), Taiho Pharmaceutical (Tokyo, Japan), VCN Biosciences (Barcelona, Spain), Biocartis (Biocartis Group, Mechelen, Belgium), Foundation Medicine (Cambridge, MA, USA), HalioDX (Marseille, France), SAS Pharmaceuticals (Delhi, India) and Roche Diagnostics outside the submitted work. Bengt Glimelius reports support from PledPharma AB for being on advisory boards. Sherif Raouf reports grants, personal fees and non-financial support from Roche, grants and personal fees from Amgen, and grants and personal fees from Merck outside the submitted work. David Farrugia reports that he received honoraria for speaking in educational events and support for meeting attendance

^{*}Corresponding author tim.iveson@uhs.nhs.uk

from Bristol-Myers Squibb, Novartis, Ipsen, Amgen, AstraZeneca and Merck. David Cunningham reports grants from 4SC (4SC AG, Planegg, Germany), AstraZeneca, Bayer, Amgen, Celgene, Clovis Oncology (Boulder, CO, USA), Eli Lilly and Company, Janssen Pharmaceuticals (Beerse, Belgium), MedImmune (Gaithersburg, MD, USA), Merck, Merrimack and Sanofi, outside the submitted work. Tamish Hickish reports grants from Pfizer, Roche, Pierre Fabre (Paris, France) and personal fees from Eli Lilly and Company during the conduct of the study. John Bridgewater reports funding from the University College London Hospitals NHS Foundation Trust/University College London Biomedical Research Centre. David Cunningham reports funding from the National Institute for Health Research Biomedical Research Centres at the Royal Marsden.

Published December 2019 DOI: 10.3310/hta23640

Plain English summary

SCOT RCT: 3-year follow-up

Health Technology Assessment 2019; Vol. 23: No. 64

DOI: 10.3310/hta23640

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Patients diagnosed with bowel cancer are likely to have surgery to remove the tumour. Patients diagnosed with a more advanced stage of the disease are then likely to be offered what is known as adjuvant chemotherapy – chemotherapy to kill any cancer cells that have already spread but cannot be seen. Adjuvant chemotherapy is usually given over 6 months using two medicines known as oxaliplatin and fluoropyrimidine. This chemotherapy has side effects of diarrhoea, nausea and vomiting, and it reduces the numbers of cells in the blood. It can also damage nerves, which causes discomfort, numbness and tingling; in some cases, this can go on for years. These side effects are more likely to develop with longer treatment. This study looked at whether or not shortening the time over which patients were given oxaliplatin and fluoropyrimidine chemotherapy reduced its effectiveness.

In this large study of over 6000 patients, half of the patients were allocated by chance to be treated for 3 months and the other half to be treated for 6 months. Reducing the time that patients had chemotherapy from 6 months to 3 months did not make the treatment less effective. When patients treated with chemotherapy over 3 months were compared with those treated over 6 months, 77% of patients in both groups were well with no detectable disease 3 years after surgery. Patients were less likely to get side effects with 3-month chemotherapy. In particular, the chance of persistent long-term nerve damage was lower, resulting in patients with 3-month chemotherapy having better health-related quality of life.

Overall, the study showed that 3-month adjuvant chemotherapy for patients with bowel cancer is as effective as 6-month adjuvant chemotherapy and causes fewer side effects.

HTA/HTA TAR

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.819

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 14/140/84. The contractual start date was in December 2015. The draft report began editorial review in August 2018 and was accepted for publication in June 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Iveson et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk